

Houlihan, CF; McGowan, CR; Dicks, S; Baguelin, M; Moore, DAJ; Mabey, D; Roberts, CH; Kumar, A; Samuel, D; Tedder, R; Glynn, JR (2017) Ebola exposure, illness experience, and Ebola antibody prevalence in international responders to the West African Ebola epidemic 2014-2016: A cross-sectional study. PLoS medicine, 14 (5). e1002300. ISSN 1549-1277 DOI: https://doi.org/10.1371/journal.pmed.1002300

Downloaded from: http://researchonline.lshtm.ac.uk/4052068/

DOI: 10.1371/journal.pmed.1002300

Usage Guidelines

 $Please \ refer \ to \ usage \ guidelines \ at \ http://researchonline.lshtm.ac.uk/policies.html \ or \ alternatively \ contact \ researchonline@lshtm.ac.uk.$ 

Available under license: http://creativecommons.org/licenses/by/2.5/



# 

**Citation:** Houlihan CF, McGowan CR, Dicks S, Baguelin M, Moore DAJ, Mabey D, et al. (2017) Ebola exposure, illness experience, and Ebola antibody prevalence in international responders to the West African Ebola epidemic 2014–2016: A cross-sectional study. PLoS Med 14(5): e1002300. https://doi.org/10.1371/journal.pmed.1002300

Academic Editor: Tom Boyles, Groote Schuur Hospital, SOUTH AFRICA

Received: December 17, 2016

Accepted: April 4, 2017

Published: May 16, 2017

**Copyright:** © 2017 Houlihan et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: A full patient-level data-set is available to appropriately qualified researchers. Access to the data is via application to an independent Data Access Committee convened by the Wellcome Trust. Applications for accesss to the data can be made via the website <u>https://www.</u> iddo.org/data-sharing/accessing-data/accessingebola-data

**Funding:** CFH and JRG received funding from the Wellcome Trust: Enhancing Research Activity in

RESEARCH ARTICLE

# Ebola exposure, illness experience, and Ebola antibody prevalence in international responders to the West African Ebola epidemic 2014–2016: A cross-sectional study

Catherine F. Houlihan<sup>1,2</sup>\*, Catherine R. McGowan<sup>3,4</sup>, Steve Dicks<sup>5,6</sup>, Marc Baguelin<sup>7,8</sup>, David A. J. Moore<sup>1</sup>, David Mabey<sup>1</sup>, Chrissy h. Roberts<sup>1</sup>, Alex Kumar<sup>9</sup>, Dhan Samuel<sup>6†</sup>, Richard Tedder<sup>5,6</sup>, Judith R. Glynn<sup>8</sup>

 Faculty of Infectious and Tropical Diseases, London School of Hygiene & Tropical Medicine, London, United Kingdom, 2 Faculty of Medical Sciences, University College London, London, United Kingdom,
 Faculty of Public Health and Policy, London School of Hygiene & Tropical Medicine, London, United Kingdom, 4 Humanitarian Public Health Technical Unit, Save the Children UK, London, United Kingdom,
 Transfusion Microbiology, National Health Service Blood and Transplant, London, United Kingdom,
 NHSBT/PHE Blood Borne Virus Unit, Serology Development Unit, Public Health England, London, United Kingdom, 7 Centre of Infectious Disease Surveillance and Control, Public Health England, London, United Kingdom, 8 Faculty of Epidemiology and Population Health, London School of Hygiene & Tropical Medicine, London, United Kingdom, 9 Department of Infection and Tropical Medicine, University Hospitals of Leicester NHS Trust, Leicester, United Kingdom

† Deceased.

\* Catherine.Houlihan@lshtm.ac.uk

# Abstract

# Background

Healthcare and other front-line workers are at particular risk of infection with Ebola virus (EBOV). Despite the large-scale deployment of international responders, few cases of Ebola virus disease have been diagnosed in this group. Since asymptomatic or pauci-symptomatic infection has been described, it is plausible that infections have occurred in health-care workers but have escaped being diagnosed. We aimed to assess the prevalence of asymptomatic or pauci-symptomatic infection, and of exposure events, among returned responders to the West African Ebola epidemic 2014–2016.

# Methods and findings

We used snowball sampling to identify responders who had returned to the UK or Ireland, and used an online consent and questionnaire to determine their exposure to EBOV and their experience of illness. Oral fluid collection devices were sent and returned by post, and samples were tested using an EBOV IgG capture assay that detects IgG to Ebola glycoprotein. Blood was collected from returnees with reactive samples for further testing. Unexposed UK controls were also recruited.

In all, 300 individuals consented, of whom 268 (89.3%) returned an oral fluid sample (OFS). The majority had worked in Sierra Leone in clinical, laboratory, research, and other roles. Fifty-three UK controls consented and provided samples using the same method.

Epidemic Situations, grant number WT-106491/Z/ 14/Z. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** I have read the journal's policy and the authors of this manuscript have the following competing interests: RT has received funding from the Wellcome Trust via the University of Liverpool, and has received non-financial support from NHSBT, as part of the Convalescent Plasma Study. All other authors have declared that no competing interests exist.

Abbreviations: DABA, double antigen bridging assay; EBOV, Ebola virus; EMT, emergency medical team; ETC, Ebola treatment centre; EVD, Ebola virus disease; HCW, healthcare worker; IQR, interquartile range; LSHTM, London School of Hygiene & Tropical Medicine; NGO, nongovernmental organisation; NOD, normalised optical density; OD, optical density; OFS, oral fluid sample; PHE, Public Health England; PPE, personal protective equipment; WASH, water, sanitation, and hygiene; WHO, World Health Organization. Of the returnees, 47 (17.5%) reported that they had had a possible EBOV exposure. Based on their free-text descriptions, using a published risk assessment method, we classified 43 (16%) as having had incidents with risk of Ebola transmission, including five intermediate-risk and one high-risk exposure. Of the returnees, 57 (21%) reported a febrile or diarrhoeal illness in West Africa or within 1 mo of return, of whom 40 (70%) were not tested at the time for EBOV infection.

Of the 268 OFSs, 266 were unreactive. Two returnees, who did not experience an illness in West Africa or on return, had OFSs that were reactive on the EBOV IgG capture assay, with similar results on plasma. One individual had no further positive test results; the other had a positive result on a double-antigen bridging assay but not on a competitive assay or on an indirect EBOV IgG ELISA. All 53 controls had non-reactive OFSs. While the participants were not a random sample of returnees, the number participating was high.

#### Conclusions

This is the first study, to our knowledge, of the prevalence of EBOV infection in international responders. More than 99% had clear negative results. Sera from two individuals had discordant results on the different assays; both were negative on the competitive assay, suggesting that prior infection was unlikely. The finding that a significant proportion experienced "near miss" exposure events, and that most of those who experienced symptoms did not get tested for EBOV at the time, suggests a need to review and standardise protocols for the management of possible exposure to EBOV, and for the management of illness, across organisations that deploy staff to outbreaks.

#### Author summary

#### Why was this study done?

• A large number of international volunteers worked in West Africa during the largest Ebola outbreak ever seen, but only a small number of international staff were diagnosed with Ebola virus disease. Since asymptomatic or unrecognised infections can occur, it is possible that some infections in international responders to the epidemic were missed.

#### What did the researchers do and find?

• We carried out an online survey and antibody testing of oral fluid samples from 268 individuals who returned to the UK and Ireland after working in West Africa during the Ebola outbreak. A high number of near miss events were reported, and less than a third of individuals who experienced illness whilst in West Africa or soon after return were tested for Ebola virus while they were unwell. Of the 268 who were tested for antibodies, two had reactive test results, but these were not reactive on further testing.

#### What do these findings mean?

- Although our study could not reach every individual who returned to the UK after working in West Africa during the Ebola outbreak, we demonstrated that asymptomatic or pauci-symptomatic Ebola virus infection was rare in international healthcare and other workers in our study who responded to the epidemic in West Africa.
- The descriptions of near miss events and the finding that many of those who experienced illness were not tested at the time suggest that protocols for the management of possible exposure to Ebola virus and for the management of illness should be reviewed and standardised across organisations that deploy staff to outbreaks.

# Introduction

In Ebola virus disease (EVD) epidemics, healthcare workers (HCWs) are disproportionately affected. In the West African 2014-2016 Ebola epidemic, the percentage of HCWs infected was 1.45% in Guinea (compared to 0.02% in the general population), 8.07% in Liberia (compared to 0.11%), and 6.85% in Sierra Leone (compared to 0.06%) [1]. The outbreak occurred in countries that have some of the lowest proportions of HCWs per head of population in the world [2]. This fact—coupled with the rapid spread of the epidemic, the high case fatality rate [3–7], and serious concerns raised by the international community—motivated the World Health Organization (WHO) to declare the outbreak as a "public health emergency of international concern" and resulted in the escalated deployment of international emergency medical teams (EMTs) to West Africa (primarily to Guinea, Liberia, and Sierra Leone) beginning in late 2014 [8]. These EMTs included water, sanitation, and hygiene (WASH) and laboratory workers, epidemiologists, engineers, logisticians, and operations staff as well as clinicians. To date, WHO reports that 40 organisations from 19 countries sent EMTs to West Africa in response to the 2014–2016 Ebola outbreak [9]; over 3,000 individuals were deployed to West Africa by the US Centers for Disease Control and Prevention alone [10]. Despite this largescale deployment, few cases of EVD were diagnosed among EMT workers: 24 were treated in Europe or the US [8].

In Kikwit, Democratic Republic of the Congo, in an Ebola outbreak in 1995, 8/402 (2%) HCWs in the outbreak area had positive serology in the absence of diagnosed EVD. Among asymptomatic household contacts of cases, the prevalence of positive serology has ranged from 1% to 46% in different studies using different assays [11–16]. More recently, serology positive for Ebola virus (EBOV) infection was found in 12.0% (11/92) of household contacts with symptoms in Sierra Leone, and 2.6% (10/389) of those with no symptoms, using a highly sensitive and specific IgG assay [17]. A recent meta-analysis estimated that 27% of all infections are asymptomatic [18], with similar findings from a "hot spot" in Sierra Leone [19]. It is therefore plausible that asymptomatic or pauci-symptomatic infections have occurred in HCWs but have escaped being diagnosed.

We describe the prevalence of antibodies to EBOV in oral fluid samples (OFSs) from individuals who returned to the UK or Ireland after responding to the West African EVD epidemic, as well as potential exposures to EBOV and experience of illness in West Africa or within 1 mo of return.

## Methods

#### Recruitment

Invitations to participate were sent to individuals known to the authors, and through organisations supporting EMT deployment involving UK-based staff, including non-governmental organisations (NGOs), UK government-affiliated institutions, and the London School of Hygiene & Tropical Medicine (LSHTM). A request was embedded in both the email and at the end of the online questionnaire to forward the email to other eligible persons—"snowball sampling" [20]. The email described the purpose of the study and the methods of data collection, and included a link to an online information sheet, consent form, and questionnaire.

The study was also advertised via posters and cards at selected meetings that eligible individuals were likely to attend (with permission from the meeting organisers). The study information and a link to the survey were also distributed via social media platforms (Facebook and Twitter).

UK control individuals with no known exposure to filoviruses were also recruited at LSHTM and through personal contacts.

An online consent procedure was employed for this study and consisted of an information sheet and consent form included at the beginning of the online survey (see <u>S2 Text</u>). The survey was accessible only after consent and was opened on 16 December 2015 and closed on 16 June 2016. Ethics approval was obtained from the LSHTM Research Ethics Committee (approval #9475). All data were stored on the LSHTM secure data server and were subject to file access logging.

#### Eligibility criteria

Eligible participants were  $\geq$ 18 y of age, had travelled to West Africa during the 2014–2016 EVD outbreak as part of the response, had never tested positive for EBOV by PCR, and had never received a filovirus vaccine.

UK controls were ineligible if they were <18 y of age, had taken immunosuppressant medication in the past year, had spent a prolonged period (>1 mo) in any country with EBOV transmission, had ever had contact with patients with exposure to or infected with filoviruses, had worked with filoviruses in a laboratory setting, or had participated in a filovirus vaccine study.

#### Questionnaire

The questionnaire asked for basic demographic information and information about returnees' deployment to West Africa, and any occupational risks or potential exposures (see S2 Text). Returnees were asked if they had experienced a febrile illness during their deployment or within 1 mo of return. For those using personal protective equipment (PPE), details were asked about the methods used for doffing. A broad definition of PPE was used but included a protective suit (with or without hood) with gloves (single or multiple pairs) and with eye protection, with or without respiratory protection.

#### Oral fluid collection kit

Oral fluid collection devices (Oracol S10, Malvern Medical Developments) were sent via the UK postal service to consenting eligible individuals. Returnees collected the OFS using a sponge massaged along the gum line for 90 s. Returnees could follow a pictorial instruction leaflet or follow a link to a video demonstration (http://tinyurl.com/lshtmsample).

In order to minimise potential delays, a request to post specimens within 24 h of collection was included. Specimens were stored at  $-80^{\circ}$  C°C within 24 h of receipt at LSHTM. Collection continued until 300 returnees had consented to participate and recruitment was slowing. All analyses were done after the end of data collection.

# Sample testing

The OFSs were tested for EBOV glycoprotein IgG using an IgG capture assay. Oral fluid EBOV glycoprotein IgG levels have been shown to be closely correlated with plasma IgG level  $(R^2 = 0.78)$  [14]. In a field study in Sierra Leone, the assay had a sensitivity of 96% (93/97 recovered patients with PCR-confirmed EVD tested positive) and a specificity of 100% (339/339 controls tested negative) [17]. Testing was carried out in two batches on 12 May 2016 and 22 June 2016, and samples received after 22 June 2016 were destroyed. Specimens were defrosted and extracted with 1 ml of transport medium. One hundred microlitres of extract was incubated in individual wells coated with anti-human IgG, along with two positive and two negative controls, before being washed. A conjugate of recombinant EBOV Zaire (Mayinga strain) glycoprotein antigen (rGPδTM, cat. 0501–016, IBT Bioservices) coupled with horseradish peroxidase antigen was added, further incubated, then washed. Bound conjugate was detected using TMB substrate, and optical density (OD) measurements were made on an ELx808 Ultra Microplate Reader (Biotek Instruments) and analysed using Kineticalc software for Windows (BioTek). The cutoff for each plate was taken as the mean of the negative controls in that plate plus 0.1 OD, derived from a standard curve method applied to samples reported elsewhere [21,22]. In order for a plate to be valid, the OD of the positive control had to be >1.0, and that of the negative control <0.1. Normalised ODs (NODs) were calculated using the ratio of the test OD to the defined test cutoff. A NOD > 1 was considered reactive.

Those with reactive OFSs were contacted, and 5 ml of blood was collected. Plasma samples were tested at Public Health England (PHE) for EBOV RNA using a real-time pan-Ebola PCR [23], and for EBOV antibodies using the IgG capture assay and, additionally, a competitive ELISA that included a labelled murine monoclonal 4G7 antibody and a double antigen bridging assay (DABA), both of which use the same recombinant glycoprotein as the IgG capture assay. Finally, plasma samples were sent to the Jenner Institute in Oxford, where they were tested using an ELISA that detects IgG to glycoprotein of EBOV Mayinga strain (amino-acids 1–649 of GenBank protein AHX24649.1), an in-house assay used to measure Ebola vaccine response [24]. Because of a limited supply of recombinant glycoprotein antigen and monoclonal antibody at PHE, no negative samples were tested using these second-line tests. The positive control in all PHE tests was a 1:200 dilution of serum from a recovered UK Ebola patient. No internal quality control was incorporated to verify the presence of adequate total immunoglobulin in the OFSs, but all samples were checked visually to ensure that the un-extracted swabs were not dry.

#### Statistical methods

Analyses were performed using STATA 14 (StataCorp).

The risk of infection in individuals was categorised as very low, low, intermediate, or high, using the stratification devised by Jacobs et al. [25], applied to the survey free-text answers from returnees, with modification to include breaches in PPE. Using this system, being physically close (<1 m) to an individual with confirmed EVD whilst not wearing PPE, or with a breach in the PPE, constituted very low risk if there was no direct physical contact, low risk if there was direct physical contact, and intermediate risk if the patient had vomiting, diarrhoea, or bleeding. Contact with body fluids, either directly from a patient or with the environment,

was considered intermediate risk if there was no exposure onto broken skin or mucous membranes or high risk if there was contact with these areas. In transcutaneous exposure (such as a needle stick injury), risk was considered high if the sharp was a freshly used hollow-bore needle and high if the sharp object had been contaminated by contact with a confirmed EVD patient or body fluids. The risk from a percutaneous injury was considered intermediate if the sharp was not known to have been directly contaminated [25].

Factors associated with risk of exposure, experiencing a febrile/diarrhoeal illness in West Africa or within 1 mo of return, and being tested for EBOV during such illness were determined using logistic regression. Factors reaching p < 0.1 in the univariate analysis were included in a multivariable model. Tests for trend were calculated using the likelihood ratio test to compare a linear model with a categorical model.

## Results

#### Participants

It was not possible to estimate the number who received the survey via email, but online consent was received from 300 individuals, all of whom completed the survey; of these, 268 (89.3%) returned an OFS and were included in the analysis. Fifty-three UK control participants consented and provided an OFS using the same procedures.

## Descriptive data

The median age for those who returned a sample was slightly older than for those who did not (39 versus 33 y, respectively; p = 0.07); all other characteristics were similar between the groups. Of the 268 returnees who returned samples, 152 (57%) were female; the median age was 36 y (interquartile range [IQR] 30–45). The majority had most recently worked in Sierra Leone (253, 94%) and had worked in PPE (230, 86%) (Table 1). The median age of the controls was 35 y (IQR 31–40), and 35 (66%) were female.

#### Contact and exposure

Possible exposure was ascertained through several different questions, and free-text responses were encouraged to describe incidents more fully. Twenty-one (8%) individuals reported having had direct physical contact with patients with confirmed or suspected EVD whilst not wearing PPE. Fifty-five (21%) answered "yes" (47) or "unsure" (8) when asked whether they had had a significant exposure or a "near miss" event. Using the free-text descriptions, exposures were allocated a level of risk of transmission (Box 1). In all, 43 (16%) individuals had potential exposures: 27 (10%) very low risk, 10 (4%) low risk, five (2%) intermediate risk, and one high risk (Fig 1).

Table 2 shows the factors associated with reporting any transmission risk compared to reporting no/unclear risk. The amount of time spent in PPE in the laboratory or the clinical red zone was associated with increased odds of experiencing an EBOV transmission risk (*p* for trend = 0.02, excluding those not wearing PPE). Returnees were more likely to report an exposure risk if they had been performing a clinical role (29% of 123) compared to a laboratory role (3% of 90) (odds ratio 13.4 [95% CI 3.8–47.0] after adjusting for time in PPE) (Table 2). After adjusting for role and days in PPE, no other factors were associated with risk. The different methods of PPE removal were associated with the role, so were not independently associated with risk.

Over one-third of returnees reported contact with recovered EVD patients (98/268, 37%). Most of these (57/98, 58%) were direct physical contact (handshaking or embracing) with

Characteristic	Number (percent) or median (IQR)		
Demographics			
Female	152 (56.7%)		
Age (years)	36 (30–45)		
Country most recently worked in			
Sierra Leone	253 (94.4%)		
Liberia	12 (4.5%)		
Guinea	3 (1.1%)		
Roles undertaken (returnees could select more than one)			
Laboratory	95		
Physician	70		
Nurse	54		
Research	37		
Management/operations	28		
Trainer	23		
Epidemiologist	19		
Community engagement/tracing	18		
WASH staff	11		
Finance	3		
Engineer	3		
Pharmacist	2		
Social worker/burial team/information technology/journalist/visitor/ logistician/nutritionist <sup>1</sup>	7		
Worked in PPE (laboratory or clinical)	233 (86.9%)		
Time in country (days)	30 (20–40)		
Previous experience			
Laboratory work with filoviruses	11		
Haemorrhagic fever outbreak	7		
Both lab and outbreak work	3		

 Table 1. Characteristics of 268 individuals who travelled to West Africa in response to the 2014–2016

 Ebola epidemic.

<sup>1</sup>One returnee reported each of these roles.

IQR, interquartile range; PPE, personal protective equipment; WASH, water, sanitation, and hygiene.

https://doi.org/10.1371/journal.pmed.1002300.t001

patients at the time of discharge from an Ebola treatment centre (ETC), which was encouraged. Others described living or working with recovered patients (9, 3%) or contact within a follow-up clinic setting (14, 5%), or they reported contact but did not provide a clear description (18, 7%). These contacts were not included as transmission risks.

#### Self-reported illness and testing

Among 268 returnees who were never known to have had EVD, 57 (21%) experienced a febrile/diarrhoeal illness either in West Africa or within 1 mo of return. <u>Table 3</u> shows the factors associated with illness and with being tested for EBOV if the returnee experienced diarrhoea and/or fever.

Clinical staff were the most likely to have experienced an illness; 29% (35 of 123) compared to 20% (11 of 55) of "other" staff and 12% (11 of 90) of laboratory staff (p < 0.01) (Table 3). After adjusting for role, no other factors were associated with illness.

Box 1. Examples of risk classification, and associated free-text descriptions from the survey, among 268 individuals who travelled to West Africa in response to the 2014–2016 Ebola epidemic

**No or unclear risk:** direct contact with local population with no known illness; torn outer glove, inner glove intact.

"Children and some adults would approach you unexpectedly and grab your arm etc. Contact with locals, they did not look ill as were mainly walking on streets or playing"

"Torn glove in isolator but remaining gloves intact and no fluid contact"

**Very low risk:** living or working with someone with undiagnosed EVD but no direct physical contact; breaches in PPE with no visible contamination or broken skin exposure

"Torn suit when caring for a sick woman who was fitting"

"Facemask dislodged"

"Person I was staying in same accommodation with became infected"

**Low risk:** direct physical contact with patient who does not have vomiting, diarrhoea, or bleeding.

"Contact with subsequently confirmed infected Medic. . .contact was symptomatic but 'dry'"

"A local colleague working at our ETU [Ebola treatment unit] came to work while symptomatic for approximately 5 days without informing us, he tested positive and died from EVD a week later"

**Intermediate risk:** direct physical contact with a patient with diarrhoea, vomiting, or bleeding; direct contact with bodily fluids from a patient with EVD.

"Cut hand on artesunate vial (through gloves) whilst in red-zone"

"Being vomited on while wearing just gloves"

"Touched mask with blood on gloves (could not breathe—had to pull mask from face)"

**High risk:** exposure of mucous membranes or broken skin to bodily fluids from an EVD patient, or sharp injury from a needle/other used on an EVD patient (not freshly used or not hollow-bore needle).

"Sharp injury with a broken vial of medication inside the red zone with dirty and contaminated gloves"



https://doi.org/10.1371/journal.pmed.1002300.g001

Amongst 57 returnees who experienced a febrile/diarrhoeal illness, 30% (17 returnees) were tested for EBOV infection at the time. One of 21 (5%) who fell ill in country had an EBOV PCR, compared to 11 of 17 (65%) who fell ill on return (odds ratio 36.7 [95% CI 3.9–244.8]). Of those who were ill and were classified as having no or unclear risk (47 returnees), 12 (25%) were tested, compared to five of the ten (50%) with some risk. The individual with a high-risk exposure did not describe an illness and was not tested. Of the five with intermediate-risk exposures, three were ill, and two of those were tested.

#### Laboratory results

The frequency of NOD values for the 268 OFSs are shown in Fig 2, alongside those from 53 controls. All the samples from UK controls and all but two of the samples from returnees were well below the cutoff (maximum NOD 0.7). The two samples that were reactive on the first test had similar results on re-testing. Plasma from these individuals was tested further, as described below.

#### Participant A

Participant A was a female laboratory worker who spent a total of 30 d in an Ebola laboratory in West Africa. She did not describe a significant exposure or febrile illness, and was not tested with PCR for EBOV infection at any point during her deployment or on return.

The OFS was reactive on three tests, with NOD 3.55, 3.27, and 3.12. In plasma, the IgG capture assay was reactive twice (NOD 4.5 and 5.4), the DABA was reactive twice (NOD 2.6 and 3.2), but the competitive assay was unreactive twice (NOD 0.44 and 0.49), and the in-house indirect ELISA test at the Jenner Institute was unreactive. The EBOV PCR on plasma was negative.

# Participant B

Participant B was a female physician who spent 40 d in the clinical red zone, did not describe a significant exposure or any illness, and was not PCR tested for EBOV infection during her deployment or on return. The OFS showed a low level of reactivity on the IgG capture assay (NOD 1.05, 1.26, and 0.88). Similar low-level reactivity was seen in the IgG capture assay of

Factor	Number (percent)	Risk of EVD transmission, <i>n</i> (percent)	Unadjusted OR (95% CI)	Adjusted OR <sup>1</sup> (95% CI)	<i>p</i> -Value
Sex					
Female	152 (56.7)	20 (13.2)	1		
Male	116 (43.3)	23 (19.8)	1.6 (0.8–3.1)		
Age group (years)					
19–30	73 (27.2)	8 (11.0)	1		
31–40	99 (36.9)	19 (19.2)	1.9 (0.8–4.7)		
41–50	45 (16.8)	8 (17.8)	1.8 (0.6–5.1)		
	51 (19.0)	8 (16.7)	1.5 (0.5–4.3)		
Country					
Sierra Leone	253 (94.4)	41 (16.2)	1		
Guinea/Liberia	15 (5.6%)	2 (13.3)	0.8 (0.2–3.7)		
Facility <sup>2</sup>					
Facility 1	72 (26.9)	16 (22.2)	1		
Facility 2	67 (25.0)	7 (10.5)	0.4 (0.2–1.1)		
Facility 3	53 (19.8)	3 (5.7)	0.2 (0.1–0.8)		
Facility 4	13 (4.9)	6 (46.2)	3.0 (0.9–10.2)		
Other	63 (23.5)	11 (17.5)	0.7 (0.3–1.7)		
Role <sup>3</sup>					<0.01
Laboratory work	90 (33.6)	3 (3.3)	1	1	
Clinical work	123 (45.9)	35 (28.5)	11.5 (3.5–38.9)	13.4 (3.8–47.0)	
Other role <sup>4</sup>	55 (20.5)	5 (9.1)	2.9 (0.7–12.7)	8.7 (1.6–47.4)	
Days in PPE <sup>5</sup>					0.02; test for trend, 0.02 <sup>6</sup>
1–25	77 (28.7)	13 (16.9)	1	1	
26–35	83 (31.0)	9 (10.8)	0.6 (0.2–1.5)	1.3 (0.5–3.6)	
36–60	38 (14.2)	9 (23.7)	1.5 (0.6–4.0)	2.3 (0.8–6.4)	
>60	22 (8.2)	9 (41.0)	3.4 (1.2–9.6)	4.7 (1.5–14.8)	
None	48 (17.9)	3 (6.3)	0.3 (0.1–0.4)	0.3 (0.1–1.4)	
Previous viral haemorrhagic fever work					
No previous work	247 (92.2)	41 (16.6)	1		
Previous work	21 (7.8)	2 (9.5)	0.5 (0.1–2.3)		
PPE removal—chlorine spray <sup>7</sup>					
No spray	98 (36.6)	7 (7.1)	1		
Chlorine spray	132 (49.3)	33 (25.0)	4.3 (1.8–10.3)		
Not applicable	38 (14.2)	3 (7.9)	1.1 (0.3–4.6)		
PPE removal—assistance <sup>7</sup>					
No assistance	108 (40.3)	13 (12.0)	1		
Assistance	122 (45.5)	27 (22.1)	2.1 (1.0-4.3)		
Not applicable	38 (14.2)	3 (7.9)	0.6 (0.2–2.3)		

Table 2. Factors associated with risk of Ebola virus disease transmission in 268 returnees who worked in West Africa during the 2014–2016 Ebola outbreak.

<sup>1</sup>Only days in PPE and laboratory work were included in the final model.

<sup>2</sup>Facility is the Ebola treatment centre or holding facility at which the returnee was predominantly based. Facilities 1–4 were the most commonly reported facilities.

<sup>3</sup>It was possible to select more than one role. Where individuals selected both laboratory and clinical red zone (n = 3), they were allocated to the role in which they reported more time in subsequent questions. "Clinical" includes nurses, doctors, and paramedics.

<sup>4</sup>Other roles include those listed in Table 1.

<sup>5</sup>Thirty-eight individuals did not spend any time in PPE. An additional ten did not provide information on the duration of time in PPE.

<sup>6</sup>Excluding those who did not wear PPE.

<sup>7</sup>On exiting the clinical red zone or laboratory. PPE removal was not included in the final model since method of removal was almost collinear with role. CI, confidence interval; EVD, Ebola virus disease; OR, odds ratio; PPE, personal protective equipment.

https://doi.org/10.1371/journal.pmed.1002300.t002

Factor	Number (percent)	Experience of illness, <i>n</i> (percent)	OR (95% CI)	<i>p</i> -Value	Tested for Ebola virus, <i>n</i> (percent)	OR (95% CI)	<i>p</i> -Value
Sex							
Female	152 (56.7)	32 (21.1)	1		10 (31.3)	1	
Male	116 (43.3)	25 (21.6)	1.0 (0.6–1.9)		7 (28.0)	0.8 (0.3–2.7)	
Age group (years)							
19–30	73 (27.2)	20 (27.4)	1		6 (30.0)	1	
31–40	99 (36.9)	21 (21.2)	0.7 (0.4–1.4)		6 (28.6)	0.9 (0.2–3.6)	
41–50	45 (16.8)	8 (17.8)	0.6 (0.2–1.4)		1 (12.5)	0.3 (0.0–3.3)	
≥51	51 (19.0)	8 (15.7)	0.5 (0.2–1.2)		4 (50.0)	2.3 (0.4–12.6)	
Role <sup>1</sup>				<0.01			
Laboratory work	90 (33.6)	11 (12.2)	1		4 (34.4)	1	
Clinical work	123 (45.9)	35 (28.5)	2.9 (1.4–6.0)		12 (34.3)	0.9 (0.2–3.8)	
Other role <sup>2</sup>	55 (20.5)	11 (20.0)	1.8 (0.7–4.5)		1 (9.1)	0.2 (0.0–1.9)	
Risk of EVD							
No risk/unclear	225 (84.0)	47 (20.9)	1		12 (25.5)	1	
Risk	43 (16.0)	10 (23.3)	1.1 (0.5–2.5)		5 (50.0)	2.9 (0.7–11.9)	
Facility <sup>3</sup>							
Facility 1	72 (26.9)	14 (19.4)	1		7 (50.0)	1	
Facility 2	67 (25.0)	18 (26.9)	1.5 (0.7–3.4)		5 (27.8)	0.4 (0.1–1.7)	
Facility 3	53 (19.8)	8 (15.1)	0.7 (0.3–1.9)		1 (12.5)	0.1 (0.0–1.5)	
Facility 4	13 (4.9)	7 (53.9)	4.8 (1.4–16.6)		3 (42.9)	0.8 (0.1–4.7)	
Other	63 (23.5)	10 (15.9)	0.8 (0.3–1.9)		1 (10.0)	0.1 (0.0–1.1)	
Illness outcome							<0.01
Illness in country	21 (36.8)				1 (4.8)	1	
Illness in UK	17 (29.8)				11 (64.7)	36.7 (3.9–244.8)	
Other cause	19 (33.3)				5 (26.3)	7.1 (0.8–68.0)	

Table 3. Factors associated with experiencing an illness in 268 returnees from the West African Ebola 2014–2016 outbreak, and factors associated with having been tested for Ebola virus in 57 who experienced an illness.

<sup>1</sup>It was possible to select more than one role. Where individuals selected both laboratory and clinical red zone (n = 3), they were allocated to the role in which they reported more time in subsequent questions. "Clinical" includes nurses, doctors, and paramedics.

<sup>2</sup>Other roles include those listed in Table 1.

<sup>3</sup>"Facility" is the Ebola treatment centre or holding facility at which the returnee was predominantly based. Facilities 1–4 were the most commonly reported facilities. Facility was not included in the adjusted model since it was almost collinear with role.

CI, confidence interval; EVD, Ebola virus disease; OR, odds ratio.

https://doi.org/10.1371/journal.pmed.1002300.t003

plasma (NOD 1.40 and 1.24). The plasma was unreactive on DABA (NOD 0.43 and 0.41), on the competitive assay (NOD 0.47 and 0.45), and on the Jenner Institute indirect ELISA. The EBOV PCR on plasma was negative.

#### Discussion

This is the first examination to our knowledge of possible undiagnosed infections with EBOV in international HCWs and other responders working during the West African 2014–2016 Ebola outbreak. More than 99% had no reactivity on the screening test, making undetected infection, at most, a very rare event. The two individuals with reactive tests for IgG to EBOV glycoprotein antigen reported no significant exposure. One of these individuals demonstrated a level of reactivity around the cutoff, illustrating non-specific reactivity, a possible finding in serological diagnostics [26]. The other individual had results that were reactive in the IgG



Fig 2. Normalised optical density results for IgG in oral fluid samples from 268 individuals who travelled to West Africa in response to the 2014–2016 Ebola epidemic, and 53 UK controls.

https://doi.org/10.1371/journal.pmed.1002300.g002

capture assay at a level seen in recovered EVD patients [17], and reactive in a DABA, but not in a competitive assay and not in a separate indirect IgG ELISA. Both the capture assay and the DABA employ exactly the same recombinant antigen (EBOV glycoprotein); therefore, concordant positivity on both could be expected. The competitive assay, however, uses the monoclonal antibody 4G7, a constituent of the proposed therapy for EVD ZMAb [27], which competes with the participant's own antibody for binding on the same recombinant EBOV glycoprotein and is expected to be inherently more specific. The result in this returnee could be explained through cross-reactivity, the presence of an antibody that binds to a single epitope on the EBOV glycoprotein used in the PHE IgG capture assay that was not present on the EBOV glycoprotein used in the Jenner Institute IgG indirect ELISA and was not produced in response to EBOV infection. Alternatively, the returnee could have been infected with a small inoculum and generated a minimal subsequent memory B cell response resulting in either a low-avidity antibody or one with a limited epitope profile. A low-titre antibody does not seem an adequate explanation since the geometric mean titres of antibody in asymptomatic undiagnosed household contacts of cases of EVD in Sierra Leone were not significantly different from those of recovered patients [17]. Since the returnee was not ever symptomatic, displayed a lack of consistent and robust reactivity on serological assays, and did not have detectable plasma viraemia, clinical follow-up was not deemed necessary, and was refused when offered.

Oral fluid EBOV IgG has been shown to have excellent concordance with plasma EBOV IgG, and the EBOV IgG capture assay has demonstrated excellent sensitivity and specificity [17]. Using this assay, seropositivity was found in 2.6% of 389 asymptomatic household contacts in Sierra Leone [17]. Using assays of different formats, 7.5% of 187 previously undiagnosed contacts were seropositive in Sierra Leone [19], and between 1% and 46% in previous outbreaks [11–13,15,16], although considerable concern has been expressed about the

specificity of the assays used to generate these data [16,28]. In our study, samples were collected from individuals at various lengths of time after their return from West Africa, but no more than 2 y. The magnitude of the antibody response has been shown to decrease over time in some vaccine studies [29] but remained detectable several years after natural infection in previous outbreaks [30]. It is possible that the sensitivity of the assay was decreased by the OFS remaining at room temperature during postage; however, this seems unlikely as antibody concentrations in oral fluid have demonstrated stability at various temperatures for up to 7 d when tested using capture assays similar to that used in our study [31].

We demonstrate no evidence of infection with EBOV in individuals who were considered to have had a risk of transmission, or in returnees who had febrile or diarrhoeal symptoms during the potential incubation time for EBOV. A worrying proportion of responders had had direct physical contact with confirmed or suspected EVD patients, or near miss or exposure events inside the red zone, many of whom met similar exposure risk criteria to those administered favipiravir as post-exposure prophylaxis [25]. None of the returnees reported having been medically evacuated or having received post-exposure prophylaxis, but this was not asked about directly.

Reporting an event classified as very low, low, intermediate, or high risk for EBOV transmission was associated with performing a clinical role rather than a laboratory role. Broken vials and laboratory PPE breaches in Ebola laboratories have been described [32]. However, laboratory work may allow greater risk prediction and protocol-guided management than working with patients, who may present with confusion during the course of EVD [6]. Regular debriefing after work in the clinical red zone or laboratory and blame-free reporting of near misses should be part of routine practice in EMT response work. Together with the further development and publication of tools through which exposure risk can be graded, staff should be given realistic advice about their level of risk to reduce confusion and avoid anxiety, while allowing those with more than minimal risk to be identified. Responses from returnees who participated in this study could be used to inform ETC design and workforce planning to prevent potential EBOV exposure. For example, reports of staff experiencing skin lacerations from broken glass vials in the red zone should lead to consideration of a complete ban on the use of glass vials in this area. Reports of dislodged or torn PPE should lead to further testing of the robustness and fit of PPE suits, careful design of ETCs so that suits are not torn by catching on doorways or corners, and confirmation that protocols exist in each ETC for the management of PPE failure whilst in the red zone.

Performing a clinical role (nurses/doctors/paramedics) was strongly associated with describing an illness, compared to laboratory work or undertaking other roles. Clinical staff wore PPE for the majority of clinical interactions, minimising the risk of acquiring any infection from EVD patients. However, caring for unwell colleagues or mixing more with West African national staff may have increased this group's exposure to transmissible pathogens to which they were immune-naïve, such as upper respiratory tract viral infections. It is concerning that a high proportion of returnees who experienced a febrile or diarrhoeal illness whilst within West Africa or within a month of return were not tested with PCR for EBOV at the time.

Being tested for EBOV infection was more strongly associated with experiencing an illness while in the UK compared to while still in West Africa. Illness in West Africa may have resulted in admission to a treatment centre where other confirmed EVD patients were being cared for, or medical evacuation to the UK. Although these would have been alarming for the individual and costly for the organisation or country of origin, earlier presentation with illness has been associated with improved survival [33,34]. The policies of NGOs, where available, were reviewed by the authors and frequently stated that EBOV should be tested for "after a

period of observation" and if symptoms have "not resolved or progressed". The pragmatic nature of these policies should be viewed in the context of the anxiety induced by testing, as well as potential exposure whilst inside a testing facility [35]. This study provides a reassuring lack of evidence of EBOV infection in individuals who had not previously been tested for EBOV, but who had had an illness consistent with mildly symptomatic EVD. Review, critique, and agreed standardisation of protocols for the management of illness in nationally and internationally recruited staff during outbreaks should be instigated.

This study included 268 individuals who returned OFSs. It was not a random sample, and it is possible that those who knew of possible exposures or who had had symptoms were particularly keen to participate. PHE conducted routine screening of all individuals who returned to the UK from West Africa at the peak of the epidemic, between 14 October 2014 and 4 December 2015. During this time, 646 individuals were designated low risk of exposure (including WASH staff, epidemiologists, health advisors, and contact tracers) or high risk of exposure (including HCWs providing EVD patient care, morgue workers, and burial teams) [36]. The use of oral fluid for testing, along with the use of a self-sampling device, is likely to have increased response rates and participation in our study [37]; it avoided the discomfort and inconvenience of blood sampling, and improved logistics for sample collection, given the geographical spread of returnees across the UK and Ireland.

By restricting the study to international staff, only 11 of whom had previous experience with filoviruses, we could evaluate the risk of contracting EBOV while working in PPE or otherwise responding to the epidemic, without the unquantifiable risk experienced by national HCWs who may have been exposed to infected family members, friends, and colleagues. Such contacts have been identified as the source of infection in the majority of HCW infections in Sierra Leone [38]. We were also able to evaluate potential transmission to individuals who worked in roles that did not predominantly involve direct work with patients with EVD, including epidemiologists, community engagement staff, researchers, and support staff such as those in finance or IT. Some of these individuals described contact with both positive and recovered cases, and considered that they had had near miss events or significant exposures. A clear correlation between level of exposure to EVD cases and their body fluids and the risk of acquiring EVD has been documented previously [17,39]. It is possible that potential transmission events or levels of risk were not captured in this study, particularly around variation in the eye protection component of PPE; EBOV had been detected in lung tissue [40], and transmission has been documented in individuals who did not recall physical contact with a confirmed case from whom they acquired infection [41].

Reporting a previously undisclosed illness during the incubation period for EVD, or not immediately reporting a near miss event, may be met with criticism or even professional disciplinary review [42]. Nevertheless, the returnees who participated in this study seemed willing to report these events. This was helped by avoiding face-to-face interaction through the use of an online consent form and questionnaire—a method that has been previously demonstrated to increase reporting of socially undesirable responses [43]—and by explicitly stating in the questionnaire that we would not identify specific treatment facilities or NGOs.

#### Conclusion

The vast majority of international responders to the EVD epidemic who participated in this study had no evidence of EBOV infection. This should provide additional evidence in support of the effectiveness of ETC infection prevention and control procedures and of PPE. In view of the lack of evidence of infection in spite of the frequency of symptoms consistent with EVD (fever and diarrhoea), and the disconcerting level of under-testing at the time, consideration

should be given to revising and standardising policies for management of illness. The level of concern expressed by returnees about potential exposure to EBOV both inside and outside red zone areas should highlight areas for improved support for national and international staff, and a need to standardise protocols for assessment and management of exposures between organisations.

# **Supporting information**

**S1 STROBE Checklist.** (DOC)

**S1 Text. EBOV IgG returnees protocol.** (DOCX)

**S2 Text. Questionnaire for returning healthcare workers.** (PDF)

#### Acknowledgments

The authors would like to thank LSHTM and PHE for providing laboratory services to support this study. We would also like to thank the Multimedia Department at LSHTM for helping to create our self-test materials. We are very grateful to Anne Koerber, Kes Stern, Magda Bondos, Dr. Colin Brown, and Dr. Landon Kuester for their assistance with this study. Additional thanks are owed to Professor Dame Anne Mills for assistance with recruitment. Finally, we are grateful to Dr. Katie Ewer and Georgina Bowyer at the Jenner Institute, University of Oxford, for testing the reactive samples from two returnees and providing advice on interpretation.

#### **Author Contributions**

Conceptualization: CFH CRM AK. Data curation: CHF CRM. Formal analysis: CFH JRG. Funding acquisition: CH JRG. Investigation: CFH CRM AK SD. Methodology: CFH CRM DS RT MB DAJM DM ChR. Project administration: CFH CRM. Resources: CFH CRM AK SD DS RT ChR. Software: CFH CRM AK SD DS RT ChR. Software: CFH CRM. Supervision: CFH JRG. Validation: CFH JRG RT. Visualization: CFH CRM JRG. Writing – original draft: CFH. Writing – review & editing: CFH CRM DAJM SD MB DM ChR AK DS RT JRG.

#### References

- Evans DK, Goldstein M, Popova A (2015) Health-care worker mortality and the legacy of the Ebola epidemic. Lancet Glob Health 3: e439–e440. https://doi.org/10.1016/S2214-109X(15)00065-0 PMID: 26163833
- 2. World Health Organization (2016) WHO global health workforce statistics. Geneva: World Health Organization. http://www.who.int/hrh/statistics/hwfstats/. Accessed 21 September 2016.
- Schieffelin JS, Shaffer JG, Goba A, Gbakie M, Gire SK, et al. (2014) Clinical illness and outcomes in patients with Ebola in Sierra Leone. N Engl J Med 371: 2092–2100. <u>https://doi.org/10.1056/</u> NEJMoa1411680 PMID: 25353969
- Bah EI, Lamah MC, Fletcher T, Jacob ST, Brett-Major DM, et al. (2015) Clinical presentation of patients with Ebola virus disease in Conakry, Guinea. N Engl J Med 372: 40–47. <u>https://doi.org/10.1056/ NEJMoa1411249 PMID: 25372658</u>
- 5. Qin E, Bi J, Zhao M, Wang Y, Guo T, et al. (2015) Clinical features of patients with Ebola virus disease in Sierra Leone. Clin Infect Dis 61: 491–495. https://doi.org/10.1093/cid/civ319 PMID: 25995207
- 6. Hunt L, Gupta-Wright A, Simms V, Tamba F, Knott V, et al. (2015) Clinical presentation, biochemical, and haematological parameters and their association with outcome in patients with Ebola virus disease: an observational cohort study. Lancet Infect Dis 15: 1292–1299. https://doi.org/10.1016/S1473-3099 (15)00144-9 PMID: 26271406
- Camacho A, Kucharski A, Aki-Sawyerr Y, White MA, Flasche S, et al. (2015) Temporal changes in Ebola transmission in Sierra Leone and implications for control requirements: a real-time modelling study. PLoS Curr 7.
- Uyeki TM, Mehta AK, Davey RT Jr, Liddell AM, Wolf T, et al. (2016) Clinical management of Ebola virus disease in the United States and Europe. N Engl J Med 374: 636–646. https://doi.org/10.1056/ NEJMoa1504874 PMID: 26886522
- 9. World Health Organization (2016) Emergency medical teams and World Health Organization. Geneva: World Health Organization. http://www.who.int/hac/techguidance/preparedness/foreign\_medical\_teams/en/. Accessed 21 September 2016.
- Centers for Disease Control and Prevention (2016) Ebola (Ebola virus disease): CDC's role. Atlanta: Centers for Disease Control and Prevention. http://www.cdc.gov/vhf/ebola/outbreaks/2014-west-africa/ what-cdc-is-doing.html. Accessed 21 September 2016.
- Clark D V, Kibuuka H, Millard M, Wakabi S, Lukwago L, et al. (2015) Long-term sequelae after Ebola virus disease in Bundibugyo, Uganda: a retrospective cohort study. Lancet Infect Dis 15: 905–912. https://doi.org/10.1016/S1473-3099(15)70152-0 PMID: 25910637
- Bertherat E, Renaut A, Nabias R, Dubreuil G, Georges-Courbot MC (1999) Leptospirosis and Ebola virus infection in five gold-panning villages in northeastern Gabon. Am J Trop Med Hyg 60: 610–615. PMID: 10348236
- Rowe AK, Bertolli J, Khan AS, Mukunu R, Muyembe-Tamfum JJ, et al. (1999) Clinical, virologic, and immunologic follow-up of convalescent Ebola hemorrhagic fever patients and their household contacts, Kikwit, Democratic Republic of the Congo. J Infect Dis 179: S28–S35. https://doi.org/10.1086/514318 PMID: 9988162
- Leroy E, Baize S, Volchkov V, Fisher-Hoch S, Georges-Courbot M-C, et al. (2000) Human asymptomatic Ebola infection and strong inflammatory response. Lancet 355: 2210–2215. PMID: 10881895
- Johnson KM, Breman JG, Burke J, Declerq R, Ghysebrechts G, et al. (1978) Ebola haemorrhagic fever in Zaire, 1976. Bull World Health Organ 56: 271–293. PMID: <u>307456</u>
- 16. Bower H, Glynn J (2016) A systematic review and meta-analysis of seroprevalence surveys of ebolavirus infection. Sci Data. 4:160133.
- Glynn J, Bower H, Johnson S, Houlihan C, Montesano C, et al. (2017) Asymptomatic infection and unrecognised Ebola virus disease: prevalence of Ebola virus sero-positivity in a large cross-sectional survey in Ebola-affected households, Sierra Leone, using a new non-invasive assay. Lancet Infect Dis. E-pub ahead of print.
- 18. Dean NE, Halloran ME, Yang Y, Longini IM (2016) Transmissibility and pathogenicity of Ebola virus: a systematic review and meta-analysis of household secondary attack rate and asymptomatic infection. Clin Infect Dis 62: 1277–1286. https://doi.org/10.1093/cid/ciw114 PMID: 26932131
- Richardson ET, Kelly JD, Barrie MB, Mesman AW, Karku S, et al. (2016) Minimally symptomatic infection in an Ebola "hotspot": a cross-sectional serosurvey. PLoS Negl Trop Dis 10: e0005087. https://doi. org/10.1371/journal.pntd.0005087 PMID: 27846221
- Biernacki P, Waldorf D (1981) Snowball sampling: problems and techniques of chain referral sampling. Sociol Methods Res 10: 141–163.

- Lambe T, Rampling T, Samuel D, Bowyer G, Ewer K, et al. (2016) Detection of vaccine induced antibodies to Ebola virus in oral fluid. Open Forum Infect Dis 3: ofw031. <u>https://doi.org/10.1093/ofid/ofw031</u> PMID: 27004234
- Frey A, Di Canzio J, Zurakowski D (1998) A statistically defined endpoint titer determination method for immunoassays. J Immunol Methods 221: 35–41. PMID: 9894896
- 23. Trombley AR, Wachter L, Garrison J, Buckley-Beason VA, Jahrling J, et al. (2010) Comprehensive panel of real-time TaqMan polymerase chain reaction assays for detection and absolute quantification of filoviruses, arenaviruses, and New World hantaviruses. Am J Trop Med Hyg 82: 954–960. https://doi.org/10.4269/ajtmh.2010.09-0636 PMID: 20439981
- Ewer K, Rampling T, Venkatraman N, Bowyer G, Wright D, et al. (2016) A monovalent chimpanzee adenovirus Ebola vaccine boosted with MVA. N Engl J Med 374: 1635–1646. https://doi.org/10.1056/ NEJMoa1411627 PMID: 25629663
- 25. Jacobs M, Aarons E, Bhagani S, Buchanan R, Cropley I, et al. (2015) Post-exposure prophylaxis against Ebola virus disease with experimental antiviral agents: a case-series of health-care workers. Lancet Infect Dis 15: 1300–1304. https://doi.org/10.1016/S1473-3099(15)00228-5 PMID: 26321189
- 26. Fierz W (2004) Basic problems of serological laboratory diagnosis. Methods Mol Med 94: 393–427. PMID: 14959841
- Zeitlin L, Whaley KJ, Olinger GG, Jacobs M, Gopal R, et al. (2016) Antibody therapeutics for Ebola virus disease. Curr Opin Virol 17: 45–49. https://doi.org/10.1016/j.coviro.2016.01.006 PMID: 26826442
- Bausch DG (2016) Sequelae after Ebola virus disease: even when it's over it's not over. Lancet Infect Dis 15: 865–866.
- Stanley DA, Honko AN, Asiedu C, Trefry JC, Lau-Kilby AW, et al. (2014) Chimpanzee adenovirus vaccine generates acute and durable protective immunity against ebolavirus challenge. Nat Med 20: 1126–1129. https://doi.org/10.1038/nm.3702 PMID: 25194571
- Natesan M, Jensen SM, Keasey SL, Kamata T, Kuehne AI, et al. (2016) Human survivors of disease outbreaks caused by Ebola or Marburg virus exhibit cross-reactive and long-lived antibody responses. Clin Vaccine Immunol 23: 717–724. https://doi.org/10.1128/CVI.00107-16 PMID: 27335383
- Morris M, Cohen B, Andrews N, Brown D (2002) Stability of total and rubella-specific IgG in oral fluid samples: the effect of time and temperature. J Immunol Methods 266: 111–116. PMID: 12133627
- Sealy TK, Erickson BR, Taboy C, Stroher U, Towner JS, et al. (2016) Laboratory response to Ebola— West Africa and United States. Morb Mortal Wkly Rep 65: 44–49.
- Crowe SJ, Maenner MJ, Kuah S, Erickson BR, Coffee M, et al. (2016) Prognostic indicators for Ebola patient survival. Emerg Infect Dis 22: 217–223. <u>https://doi.org/10.3201/eid2202.151250</u> PMID: 26812579
- Rosello A, Mossoko M, Flasche S, Van Hoek AJ, Mbala P, et al. (2015) Ebola virus disease in the Democratic Republic of the Congo, 1976–2014. Elife 4: e09015. <u>https://doi.org/10.7554/eLife.09015</u> PMID: 26525597
- 35. Arkell P, Youkee D, Brown CS, Kamara A, Kamara TB, et al. (2017) Quantifying the risk of nosocomial infection within Ebola Holding Units: a retrospective cohort study of negative patients discharged from five Ebola Holding Units in Western Area, Sierra Leone. Trop Med Int Heal 22: 32–40.
- Public Health England (2016) Ebola virus disease, activity summary. Monthly summary 14: 07/01/2016. London: Public Health England. https://www.gov.uk/government/uploads/system/uploads/attachment\_ data/file/492221/Ebola\_Activity\_Summary\_07\_01\_16.pdf. Accessed 12 April 2017.
- Virtanen A, Nieminen P, Luostarinen T, Anttila A (2011) Self-sample HPV tests as an intervention for nonattendees of cervical cancer screening in Finland: a randomized trial. Cancer Epidemiol Biomarkers Prev 20: 1960–1969. https://doi.org/10.1158/1055-9965.EPI-11-0307 PMID: 21752985
- Olu O, Kargbo B, Kamara S, Wurie AH, Amone J, et al. (2015) Epidemiology of Ebola virus disease transmission among health care workers in Sierra Leone, May to December 2014: a retrospective descriptive study. BMC Infect Dis 15: 416. https://doi.org/10.1186/s12879-015-1166-7 PMID: 26464285
- Bower H, Johnson S, Bangura MS, Kamara AJ, Kamara O, et al. (2016) Attack rates for Ebola virus disease in Ebola-affected households, Sierra Leone. Emerg Infect Dis 22: 1403–1411. <a href="https://doi.org/10.3201/eid2208.160163">https://doi.org/10.3201/eid2208.160163</a> PMID: 27144428
- Biava M, Caglioti C, Bordi L, Castilletti C, Colavita F, et al. (2017) Detection of viral RNA in tissues following plasma clearance from an Ebola virus infected patient. PLOS Pathog 13: e1006065. <u>https://doi.org/10.1371/journal.ppat.1006065</u> PMID: 28056096
- Roels TH, Bloom AS, Buffington J, Muhungu GL, Mac Kenzie WR, et al. (1999) Ebola hemorrhagic fever, Kikwit, Democratic Republic of the Congo, 1995: risk factors for patients without a reported exposure. J Infect Dis 179: S92–S97. https://doi.org/10.1086/514286 PMID: 9988170

- 42. Ebola nurse Pauline Cafferkey accused of concealing high temperature (2016). Nurs Stand 30: 8.
- **43.** Gnambs T, Kaspar K (2015) Disclosure of sensitive behaviors across self-administered survey modes: a meta-analysis. Behav Res Methods 47: 1237–1259. <u>https://doi.org/10.3758/s13428-014-0533-4</u> PMID: 25410404